Author's response to reviews

**Title:** Using Intervention Mapping to Develop and Adapt a Secondary Stroke Prevention Program in Veterans Administration Medical Centers

**Authors:**

Arlene A Schmid (arlene.schmid@va.gov)
Jane Andersen (jane.andersen@va.gov)
Thomas Kent (thomas.kent@va.gov)
Linda S Williams (linda.williams6@va.gov)
Teresa M Damush (teresa.damush@va.gov)

**Version:** 5  **Date:** 31 August 2010

Author's response to reviews:

Re: Manuscript “Using Intervention Mapping to Develop and Adapt a Secondary Stroke Prevention Program in Veterans Administration Medical Centers” MS number 1829282053285645

August 31, 2010

Thank you for the opportunity to resubmit our original manuscript, “Using Intervention Mapping to Develop and Adapt a Secondary Stroke Prevention Program in Veterans Administration Medical Centers”, for consideration for publication in the Implementation Science. This research has not been previously published.

We have made no additional changes since our last submission. We have only accepted the track changes and deleted the highlighting as requested.

All research described in this manuscript was approved by the Indiana University Institutional Review Board and the VA Research and Development Committee. Dr. Damush and this research was supported by the Department of Veteran Affairs Health Services Research & Development Service (Merit IIR 05-297-2). Dr. Schmid was supported by VA RR&D funding CDA D6174W to Dr. Schmid.

We have no disclosures to make.

Running Head: Intervention Mapping for Secondary Stroke Prevention

Word count, total: 8,441

We look forward to your comments. Thank you in advance for your time.

Please contact Arlene Schmid, Ph.D. with questions:

Sincerely,

Arlene A. Schmid, PhD, OTR
Rehabilitation Research Scientist
Roudebush VA Medical Center
1481 W. 10th Street, 11H
Indianapolis, Indiana 46202-5199

and

Assistant Professor, Department of Occupational Therapy
Indiana University- Purdue University at Indianapolis
(office) 317.988.3480
arlene.schmid@va.gov